Author’s response to reviews

Title: Association between migraine and suicidal behavior among Ethiopian adults

Authors:

Hanna Berhane (hannayaciph@gmail.com)
Bethannie Jamerson-Dowlen (bethanniejamerson@yahoo.com)
Lauren Friedman (lfriedm@hsph.harvard.edu)
Yemane Berhane (yemaneberhane@gmail.com)
Michelle Williams (mawilliams@hsph.harvard.edu)
Bizu Gelaye (bgelaye@hsph.harvard.edu)

Version: 2 Date: 29 Jan 2018

Author’s response to reviews:

January 29, 2018

Dear Editor,

BMC Psychiatry

Re: BBPSY-D-17-00589 - Association between Migraine and Suicidal Behaviors in Ethiopian Adults

Dear Editor,

Thank you for your attention to our manuscript.

We have revised our manuscript in light to the comments and suggestions provided.

Bellow we provide a point-by-point response to each comment and suggestion provided by the reviewers.

Thank you again for providing us the opportunity to revise and improve our manuscript.

Sincerely,
Comments:

1. Please include the email addresses of all authors on the title page of your manuscript.

- We have included email addresses of all authors next to their respective names.

2. Under your Ethics approval heading in the Declarations section, please clarify whether informed consent obtained from participants was written or verbal.

- We have included a line in the Ethics approval and consent to participate section indicating written informed consent were obtained prior to data collection (Line 247-248).

3. Please upload English language versions of all questionnaire/s developed specifically for use in this study as supplementary files. If the questionnaires are not yours or are already published by you, please supply references and/or links to them.

- We used the depression module of the Composite International Diagnostic Interview (CIDI) 3.0 to identify suicide behavior. The CIDI, developed by the World Health Organization (WHO) is widely used as a research diagnostic interview available on the following website: https://www.hcp.med.harvard.edu/wmhcdi/